Author's response to reviews

Title: Effect of Body Mass Index on pregnancy outcomes in nulliparous women delivering singleton babies

Authors:

Sohinee Bhattacharya (sohinee.bhattacharya@abdn.ac.uk)
Doris M Campbell (d.m.campbell@abdn.ac.uk)
William A Liston (Liston@btinternet.com)
Siladitya Bhattacharya (s.bhattacharya@abdn.ac.uk)

Version: 3 Date: 6 July 2007

Author's response to reviews:

As no further changes were requested by the reviewer, the manuscript is unchanged. In response to the editors' request, three further items have been incorporated:

1) Details of ethical approval have been added at the beginning of the methods section.

2) Contribution of each author is detailed in the section "authors' contributions" at the end of the manuscript.

3) Declaration of conflict of interest is now included in the section "competing interests" at the end of the manuscript.